

Dr James P Beirne PhD MRCOG

Subspeciality Trainee & Honorary Clinical Lecturer in Gynaecological Oncology

The Patrick G Johnston Centre for Cancer Research  
Queen's University Belfast  
97 Lisburn Road Belfast  
BT9 7AE

Tel: 07725 886988  
e-Mail: [j.beirne@qub.ac.uk](mailto:j.beirne@qub.ac.uk)

23/07/2020

Jia-Ping Yan  
Science Editor Office  
Baishideng Publishing Group Inc  
7041 Koll Center Parkway, Suite 160  
Pleasanton, CA 94566  
USA

Dear Jia-Ping,

Thank-you for giving me the opportunity to revise our manuscript. We have taken the reviewers and editorial comments on board. We have reduced the references by around 60 and included a number of key points tables to highlight important areas throughout the text.

Please contact me if there are any further queries.

Yours sincerely,

James P Beirne.